Author’s response to reviews

Title: Frequency of foot ulcers in people with type 2 diabetes, presenting to specialist diabetes clinic at a tertiary care hospital, Lahore, Pakistan”.

Authors:

Bilal Younus (bilalbin@gmail.com)

Adeela Shahid (dradeela71@gmail.com)

Rozina Arshad (rozina979@gmail.com)

Saima Khursheed (dr.cyma@yahoo.com)

Muhammad Ahmad (mlashari7@gmail.com)

Haroon Yousaf (doctharoorn@gmail.com)

Version: 3  Date: 16 Jul 2018

Author’s response to reviews:

Comments:

1. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Abbreviations

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

RESPONSE: Declaration section added at the end of conclusion and before references on page no 14.
2. Please provide three to ten keywords representing the main content of the article at the end of the Abstract

RESPONSE: Key words added at the end of abstract on page no 2.

3. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section.

RESPONSE: full initials added in author’s contribution at page no 16.

4. Please ensure to cite the poster abstract in your manuscript found at: https://onlinelibrary.wiley.com/doi/epdf/10.1111/dme.34_13048

RESPONSE: Citation added at page no 13, reference no 32